Author’s response to reviews

Title: Combined use of high doses of vasopressin and corticosteroids in chron's disease patient with refractory septic shock after intestinal perforation: a case report

Authors:

Salvatore Notaro (notarosal@gmail.com)
Marcello Sorrentino (marcello.sorrentino@tiscali.it)
Aniello Ruocco (achille1979@hotmail.it)
Annalisa Notaro (annalisa.notaro@hotmail.it)
Antonio Corcione (corcio.ant@libero.it)
Patrizia Murino (patrizia.murino@ospedalemonaldi.it)
Eugenio Piscitelli (Eugenio.Piscitelli@libero.it)
Marianna Tamborino (mariannatamborino@libero.it)

Version: 1 Date: 20 May 2017

Author’s response to reviews:

Dear authors,

before considering your manuscript for peer review process, I would like to ask you to check carefully your submission for misspells and errors - it is obvious that you report a case of Crohn’s disease - but in your manuscript title and elsewhere in the text you describe it as a "chron’s disease".

answer: thanks i have correct "chron’s"" to Crohn’s disease

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing
service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES-HS) and American Journal Experts (http://bit.ly/AJE-HS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

Answer : Thanks for the suggestion, we revised the article language, if other language reviews are needed please advise us.